Author's response to reviews

Title: The impact of transportation infrastructure on bicycling injuries and crashes: a review of the literature

Authors:

Conor CO Reynolds (c.reynolds@ires.ubc.ca)
M Anne Harris (aharris7@interchange.ubc.ca)
Kay Teschke (kay.teschke@ubc.ca)
Peter A Cripton (cripton@mech.ubc.ca)
Meghan Winters (mwinters@interchange.ubc.ca)

Version: 5 Date: 30 September 2009

Author's response to reviews: see over
Re: Manuscript #: 2000166937289161
The impact of transportation infrastructure on bicycling injuries and crashes: A review of the literature
by Conor C. O. Reynolds, M. Anne Harris, Kay Teschke, Peter A. Cripton, and Meghan Winters

We are pleased to submit the final revision of the above-named manuscript (with requested formatting changes) to *Environmental Health*.

The following gives details of the formatting change made:

- Reference 34 – As requested, this citation has been removed from the reference list at the end of the manuscript and inserted into the text as follows (page 7, first paragraph):
  “...concern about safety is one of the most frequently cited deterrents [30-34, and Winters M, Davidson G, Kao D, Teschke K: Motivators and deterrents of bicycling: factors influencing decisions to ride, submitted].”

Please do not hesitate to contact us if you have any questions regarding the revised manuscript.

Yours sincerely,

Conor Reynolds